Author’s response to reviews

Title: Feasibility and safety of setting up a donor breastmilk bank in a neonatal prem unit in a resource limited setting: An observational, longitudinal cohort study.

Authors:

Irene Coutsoudis (coutsoudis@gmail.com)
Miriam Adhikari (adhikari@ukzn.ac.za)
Nadia Nair (nairn@ukzn.ac.za)
Anna Coutsoudis (coutsoud@ukzn.ac.za)

Version: 4 Date: 19 May 2011

Author’s response to reviews:

Dear Editors

We have made the following edits as per your suggestions:

1. we have rechecked the abstract in the submission system and have edited it so that it is identical to the abstract in the manuscript.

2. We have removed the visible vertical lines from the tables

3. we have used authors’ initials in the email list.

4. we have re-checked for typo errors.

Thank you

Kind regards

Irene Coutsoudis